Author's response to reviews

Title: Evaluation of the effects of 80% methanolic leaf extract of Caylusea abyssinica (fresen.) fisch. & mey on glucose handling in normal, glucose loaded and diabetic rodents

Authors:

Wondmagegn Tamiru (mail2wondm@gmail.com)
Ephrem Engidawork (ephrem@phar.aau.edu.et)
Kaleab Asres (kasres@gmail.com)

Version: 3 Date: 21 March 2012

Author's response to reviews: see over
Dear Editor,

We profusely apologize for not including the ethical issue in the MS. It was an oversight and thanks for your reminder we have now included a statement that describes procedures followed in the handling and care of animals as well as the approval of the protocol by the School IRB. Now the MS appears to be complete and we request your editorial hand so that it could be published in your esteemed journal.

Kind Regards.

Ephrem Engidawork (PhD)